Author's response to reviews

Title: Assessing awareness of colorectal cancer symptoms: Measure development and results from a population survey in the UK

Authors:

Emily Power (emily.power@cancer.org.uk)
Alice Simon (alice.simon@ucl.ac.uk)
Dorota Juszczyk (D.Juszczyk@bath.ac.uk)
Sara Hiom (sara.hiom@cancer.org.uk)
Jane Wardle (j.wardle@ucl.ac.uk)

Version: 4 Date: 19 July 2011

Author's response to reviews: see over
18th July 2011

Dear Dr Chap,

RE: 3659665214765637 - Assessing awareness of colorectal cancer symptoms: Measure development and results from a population survey in the UK.

Thank you for sending through formatting guidance to help us prepare our manuscript for publication. We have listed the changes we have made to the manuscript below.

Affiliations: Please write the country names.

We have added 'UK' to the first affiliation so that all affiliations now have a country.

Footnotes/Appendices: Please reformat and move to a `Endnotes¿ section directly before the references. All points should be referenced with either a number or a letter.

We have inserted a section called 'Endnotes' before the References which includes all the footnotes referenced in the text.

References: reference 20, is missing from the reference list please provide details whilst ensuring references are still cited in consecutive order.

We have reformatted the references list so that reference 20 is now visible.

References: Please provide functioning link to http://www.nhs.uk/conditions/cancer-of-the-colon-rectum-or bowel/pages/keymessages.aspx

We have replaced this link.

Tables: Every table provided at the end of the manuscript file must be referred to in the text. Table 3 is not cited.

Table 3 is now cited in the text.

Email address: please include email addresses for all the authors on the title page, using the following format:

AB: abcd@institution.ac.uk
EF: efgh@generic.co.uk
IJ: IJKL@corporation.com

We have now formatted the email address list on the title page so that author names appear as initials.

Highlighting/tracking: Please remove all highlighting/tracking from the manuscript.

All highlighting and tracked changes have been removed from the manuscript.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
We have checked through the manuscript for any further typos and amended where necessary. We believe that the manuscript is now ready for publication.

Yours sincerely,

Emily Power